

LUND UNIVERSITY

In Search of Informal Caregivers in HRI: Situating and Disentangling Practices concerning Robots for Care within the HRI Conference

Tangueray, Laetitia; Larsson, Stefan; Winkle, Katie

Published in: International Journal of Social Robotics

2024

Document Version: Early version, also known as pre-print

Link to publication

Citation for published version (APA): Tanqueray, L., Larsson, S., & Winkle, K. (2024). In Search of Informal Caregivers in HRI: Situating and Disentangling Practices concerning Robots for Care within the HRI Conference. Manuscript submitted for publication.

Total number of authors: 3

General rights

Unless other specific re-use rights are stated the following general rights apply:

- Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the
- legal requirements associated with these rights

· Users may download and print one copy of any publication from the public portal for the purpose of private study You may freely distribute the URL identifying the publication in the public portal

Read more about Creative commons licenses: https://creativecommons.org/licenses/

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

LUND UNIVERSITY

PO Box 117 221 00 Lund +46 46-222 00 00

In Search of Informal Caregivers in HRI: Situating and Disentangling Practices concerning Robots for Care within the HRI Conference

Laetitia Tanqueray^{1*}, Stefan Larsson¹ and Katie Winkle²

^{1*}Department of Technology and Society, Lund University, Lund, Sweden.

²Department of Information Technology, Uppsala University, Uppsala, Sweden.

*Corresponding author(s). E-mail(s): laetitia.tanqueray@lth.lu.se; Contributing authors: stefan.larsson@lth.lu.se; katie.winkle@it.uu.se;

Abstract

Despite the acknowledged significance of informal caregivers in health care generally, it remains unclear whether and to what extent the HRI community accounts for informal caregivers. Accordingly, this article provides a systematic review of 52 articles, published within the ACM/IEEE International Conference on Human-Robot Interaction (2006-2023), to situate informal caregivers in health care within the Human-Robot Interaction (HRI) community. This study relies on a theoretical lens - deriving from Feminist HRI and Social Justice-Oriented Design - and qualitative methods to showcase four dimensions on how informal caregivers are, or not, being investigated in comparisons to other stakeholders within health care in HRI. These dimensions focus on (1) how the study is motivated, (2) who the study includes, (3) the methods used in order to answer this aim, and (4) how much caregivers are accounted for. The results point to informal caregivers only being participants in eight studies, yet are not mentioned in 15 studies, which does not echo their societal importance and our reliance in HRI on that group. We discuss our findings and provide five high-level recommendations based to account more for informal caregivers in HRI.

Keywords: HRI, Health Care, Informal Caregivers, Critical Systematic Literature Review, Social Justice, Feminist HRI

1 Introduction: Informal Care and HRI

Informal caregivers have been named by the United Nations as "co-producers of care services" [1], as they are the ones to "help prevent or delay the need for institutionalisation of people in need of care or support and are enabling them to remain living at home" [1, p.2]. In practice, this means that it is estimated that 80% of all care within the European Union (EU) is provided by informal caregivers [2], and although the exact number is not known around the world, it is most often informal caregivers providing the bulk of the care [3]. This can be explained in part by Western nations "shifting responsibility of care from the state to the voluntary and informal sectors" [4, p.2], meaning that care is being deinstitutionalised and informal caregivers are being more and more relied on. Looking more closely at the figures, in the USA, it was estimated that around 21% of the population were informal caregivers (with around 5.7%of those caregivers being under 18 years of age) [5], a similar number to the EU [6], Israel and Canada [7]. This number is slightly lower in places such as India and New Zealand (around 10%), and 5% in Japan, possibly due to varying definitions of what an informal caregiver is/does [7]. Carers UK estimates that two out of three people in the UK provide informal care at some point in their life [8]. Evidently, informal caregivers are crucial to real-world care delivery, and ought hence be considered within humanrobot interaction (HRI) research relating to care. With this review, we seek to situate informal caregivers in HRI studies and disentangle them from other key stakeholders within health care, namely: care receivers, formal caregivers, control groups, and other experts. We mostly focus on providing comparisons between informal caregivers and formal caregivers.

We take our working definition for "informal caregiver" from that of Royal College of General Practioners [9, p.1] as:

A person of any age, adult or child, who provides unpaid support to a partner, child, relative, or friend who couldn't manage to live independently or whose health or wellbeing would deteriorate without this help. This could be due to frailty, disability or serious health condition, mental ill health or substance misuse.

We note that whilst informal carers are unpaid, the work they do is very demanding – both physically and emotionally [5, 10, 11]. Furthermore, our definition of health care goes beyond that of institutions, such as hospitals or care homes and formal diagnoses of particular diseases or medical (in)capacity, which is why we use a space between 'health' and 'care'. Instead, our definition encapsulates providing care for someone to support their healthy living. Such care might relate to overall well-being, preventative health care, temporary support following for example an injury or longer-term support associated with aging and/or long term health issues. We exclude general childcare, but include situations of informal (health) care provided by parents/guardians in line with the definition given above ("parent carers" being a recognised group within the informal care literature [12]).

The main aim of this study is to identify and synthesise how informal caregivers are included and/or positioned, if at all, in HRI studies on health care. We do so by comparing and contrasting works which do and do not specifically target/include informal caregivers compared to formal caregivers and care recipients. We pursue this

by conducting a systematic literature review on articles published at the ACM/IEEE International Conference on Human-Robot Interaction. In line with previous reviews that have utilised the HRI conference proceeding as way of commenting on HRI practice [13–16], we suggest that the highly selective nature of the conference (acceptance rate typically 25% or less) suggests work published at the conference is considered high quality by the community. We would also suggest that methodologies and practices showcased at the conference are therefore likely to be seen as good practice and hence propagate through the field more broadly.

This comparison is led by a theoretical lens, namely from Social Justice-Orientated Design [17] and Feminist HRI [18]. This lens allows us to reflect and specifically focus on (1) how the study is motivated, (2) who the study includes, (3) the methods used in order to answer this aim, and (4) how much caregivers are accounted for. Our results indicate that informal caregivers are not often being considered in studies, and when they are considered, they frequently have secondary, or even tertiary, roles. Based on our results, and also drawing on additional literature beyond our article sample set, we reflect on the alignment between the role of informal caregivers in society, and those in HRI studies. This then allows us to provide high-level recommendations.

To create a foundation for the analysis, section 2 provides more insight on informal caregivers in general. The following section, section 3, introduces relevant literature; section 4 provides the theoretical lens relied on to conceptualise the study, then section 5 introduces the methodology followed to conduct the systematic literature review. The results section, section 6, showcases the four dimensions we analysed. These findings are then discussed and reflected on in section 7. This leads to five recommendations for why and how informal caregivers could be better included in HRI research, as well as a conclusion in section 8.

2 Background on Informal Caregivers

In this section, we present four key takeaways from the wider literature on informal caregivers to demonstrate the complexity and nuances of their situation, before we clarify what has, and has not, received attention in HRI with regards to informal caregiving.

Firstly, informal caregivers exist all over the world, there are issues in identifying informal caregivers. Carduff et al. [19] discuss informal caregivers' difficulty in identifying as caregivers; these include (1) becoming a 'carer' is often a gradual process, and thus they prefer to see themselves in their relational role (i.e. spouse, child of, friend), (2) care roles can be all-encompassing, rendering the caregiver unable to access relevant services, and (3) who should be identifying them – should it be the medical institution for example.

Secondly, there is a considerable amount of research pointing to the negative impact of caring can have on the informal caregiver with respect to their own overall wellbeing [10]. That said, informal caregivers are not a homogeneous group and all have different lived experiences in this regard. Hegelaar et al. [20] undertook a scoping review to demonstrate how utilising an intersectional perspective in research on

informal caregivers would show the various factors which will influence informal caregivers experiences, such as age, sex, relation to care recipient for example, and would allow better tailored solutions for caregiving. For example, within the realm of autism, Papadopoulos et al. [21] demonstrate the role of stigma within autism, which has huge (often unspoken) impacts on informal caregivers' own mental health. Another example is young carers – children caring for a relative – whose own needs are often overlooked by society despite their vital role in the family [22].

Thirdly, even if the care receiver moves into a care facility, informal caregivers will still be providing "considerable front-line work" [23, p.105]. These findings the invisible work informal caregivers often carry out. This becomes especially important when considering covid-19. Indeed, it was found that informal caregivers who were already struggling with their care responsibilities experienced the greatest difficulties during the covid-19 restrictions [24]. Furthermore, social isolation, loneliness and care burden increased for informal caregivers during covid-19 [3].

Finally, informal caregivers have been reported to be overloaded by information even as their own informational needs go unmet. For example, they may get a lot of information at a time of diagnosis, but this often omits specific information that they would like to have [25].

3 Relevant Literature

3.1 Human-Machine Interaction and Informal Caregivers

Robots are not the first digital technologies to be posited for and/or utilised by informal caregivers. LaValley et al. [26] investigated (technology-facilitated) strategies used between family caregivers and older adults for medication management. The participants pointed out that they were reluctant to use new, targeted digital technologies (such as smart phone apps) because of the time it requires for the carer to understand how to use it and then show the care recipient how/when they should use it. Instead, some speak of the makeshift technology they create themselves. Shaffer et al. [27] used the USA's Health Information National Trends Survey in order to assess age-based variations in caregivers' internet use, both generally and for seeking healthrelated information. Although the older informal caregivers were less likely to use the internet generally, over 67% of those who did used it to access health information about others. This shows that the internet is being used as a platform by informal caregivers for their role as carers. Additionally, Messina et al. [28] reviewed 'iSupport', an evidence-based online intervention developed by the World Health Organisation, in order to provide support and self-learning tools to informal caregivers of people living with dementia. However it was found that informal caregivers were reluctant to use it, as it created a "high level of burden; sense of duty; fear of being misunderstood by others; and difficulty in reaching information" [28]. These studies demonstrate that technology is being used and/or appropriated by informal care givers, but also suggest a need to better consider informal caregivers' needs and requirements when it comes to targeted intervention design.

Within HCI, these issues have already received some attention. In 2014, Schinkinger and Tellioğlu [29] published "Design implications to systems supporting informal caregivers' daily life" to bring awareness of informal caregivers and the importance of designing for them, as they will also be the ones to adapt to the new technology targeted at care receivers. Schinkinger and Tellioğlu point out that this technology can directly help informal caregivers in order to communicate with professional care services, be part of self-help groups or also monitor among other applications [29, p.349]. Since 2023, there is an increasing focus from the HCI community on informal caregivers: Smriti [30] published an extended abstract to show how they will bring informal caregivers forth into the design of artificial intelligence (AI) systems that are best suited to help them in their care role; and there is an ongoing scoping review is underway to evaluate IT applications for informal caregivers [31].

3.2 Human-Robot Interaction in Care

On a general level, current HRI research for care points to health professionals (including doctors, nurses, carers) perceiving the usefulness of social robots (see e.g. [32]), as well as elderly people (see e.g. [33]), children (see e.g. [34]) and even pets [35]. Some HRI studies specifically invite domain experts and potential users to identify how exactly robots could be helpful in the care context, with participants often able to identify possibilities that they might be enthusiastic about [36].

Literature reviews in this field have focused on, for example, what types of robotics were being researched and for what care applications, which includes social robots [37], as well as how social robots are used to help end-users within elderly care [38] or within children's mental health care [39]. Literature reviews from critical perspectives have focused on elderly care [40–42], to showcase the political nature of bringing robots into care: the power dynamic at play results in tech solutions being prioritised, rather than the societal challenges at hand.

3.2.1 HRI and Informal Caregivers

Since this article only reports on selected articles (sampling only from those published at the ACM/IEEE International Conference on Human-Robot Interaction), it is relevant to mention other HRI studies which have included informal caregivers. Amabili et al. [43] looked at the use of eWare, composed of a lifestyle-monitoring technology and a social robot with the aim of reducing stress and improving quality of life for both informal caregivers are well as people living with dementia. They report that "the impact of the system in reducing the caregivers' burden needs to be deeply investigated" [43, p.1]. With their CARESSES project, Papadopoulous et al. [21, 44] test and evaluate culturally competent socially assistive robots among older adults, describing in one study how they involved informal caregivers [44], although the findings on their involvement are unclear. Laban et al. [45] investigated the deployment of a social robot to elicit self-disclosure for informal caregivers over five weeks. The preliminary findings showed a positive trend, where informal caregivers shared more and more about their feelings over time. This demonstrates the possibility for robots to help alleviate the care burden through supporting the informal caregiver.

3.3 Critical Perspectives within Human-Robot Interaction

Researchers within HRI have identified that the demand for robots in care is seemingly not bottom-up [40, 46]. In other words, care organisations, or families, may not be the ones asking for technological solutions to help. Rather, the demand is more from institutions, embedded within allegedly *neutral* discourse emanating from governments and international organisations [40, 41]. This allegedly neutral discourse presents alarming narratives about seemingly "inevitable problems" regarding changing demographics, especially with regards to the increase in ageing population and shortage in skilled workers in care (for example [47, 48]), issues which indeed require innovative approaches. Yet those statistics do not include for example the rising struggles for care workers in part due to budget cuts in the sector [49], or the policies framing the "need" for robots in care [42], nor do they fully acknowledge informal caregivers. As a consequence, this usually results in technologies benefiting only a certain type of user – this can be within care for example, where elderly care is often the object of literature reviews within HRI (for eg [33], critical (for eg [40, 41]) and medical (for eg [38, 50]) articles. A number of recent studies have called attention to the ways in which HRI research can reproduce a (sometimes harmful) status quo, e.g. with respect to gender stereotypes [51, 52], including within care [53], and call for the HRI community to consider not only how we, the HRI community, might avoid this, but actually the role it might/ought to have in *challenging* the status quo.

Building on from this, a previous literature review considering the HRI conference proceedings demonstrated that the 'H' in HRI often represents certain homogeneous groups, calling for increased efforts to diversify research participation and make studies more accessible to a broader range of participants [16]. This is somewhat echoed in another literature review at the demographic looked at in HRI studies [54], where Seaborn, Barbareschi, and Chandra use the acronym "WEIRD" (acronym for Western, Educated, Industrial, Rich, and Democratic) to showcase the population primarily drawn on in HRI studies.

4 Theoretical Lens: Social Justice and Feminist HRI

The scope of this paper is to understand how researchers are including informal caregivers in comparison to other groups within HRI health care research at the HRI Conference. Accordingly, we map out various aspects that we believe to be key considerations regarding how various stakeholders are included within HRI health care studies. Ultimately this is where the technology has the potential to preserve or challenge current practices. Said differently, what knowledge is produced about how to use technology might overlook stakeholders which are already omitted. Accordingly, we borrow the concepts "designing for recognition" and "designing for accountability" from HCI scholars, Dombrowski, Harmon and Fox [17]'s Social Justice-Oriented Design, and mould them to our research. We also draw on the concepts of power and subject-positioning relations from Feminist HRI, by Winkle et al. [18].

Dombrowski, Harmon and Fox [17] use their starting point to demonstrate the need for social justice in HCI by pointing out that research projects examine or intervene in large scale issues – such as health care – which require scholars to engage in (or

against) state and/or personal level politics. Therefore, although design is ultimately, they argue, about creating change, those designs will create "new technological objects [which will] afford new practices, social habits, and ways of living and interacting" [17, p.1]. Thus, by explicitly engaging in social justice, researchers and designers can facilitate more equitable change.

Turning to the definition of social justice-orientated design, Dombrowski, Harmon and Fox [17] rely on Lötter's conceptualisation of social justice as a multi-dimensional concept, whereby "at its core, Lötter's conception of social justice centers on the social concerns, obligations, and ethical commitments created through social interactions" (p.3). Therefore the very notion of social justice is a continuously evolving concept, and offers, based off Lötter's conceptualisation, six dimensions: transformation, recognition, reciprocity, enablement, distribution, and accountability. These dimensions demonstrate design strategies to create an engagement with difficult, often political, questions that designers should confront. This should be achieved from a systemic perspective rather than an individual level. For the purpose of this literature review, we borrow two dimensions: recognition and accountability.

Recognition is a notion that focuses on identifying practices as well as identifying people impacted by a certain phenomenon [17, p.6] – here HRI for health care. To do so, we analyse collected articles within specific parameters to understand how HRI researchers articulate and frame research problems within HRI for health care, and who is included when answering the research question. By explicitly and systematically recognising research practices, we address social– and in turn research– challenges that might have previously been under-recognised. Therefore recognition is about focusing and identifying (possibly unjust, according to [17]) practices, whilst also showing the complexities of one phenomena – here stakeholders in health care.

The next dimension is *accountability*, which includes holding responsible, but unlike the rest of their definition of this dimension we do not sanction. Accountability in this review is not veering the analysis, but a consequence, as we are holding up a mirror to the HRI community on how research is conducted in HRI for health care. Although it is limited to the HRI Conference, we are reflecting and confronting current practices. Our hope here is that by holding HRI community accountable, we allow a reflection of our own practices, so that as researchers we can leverage on our power to decide who is part of our studies and how it could be possible to create more holistic research designs within HRI in health care.

Finally, those dimensions, and generally social justice, have to also account for power. Here, we use Winkle et al.'s [18] conceptualisation from a feminist HRI perspective. A HRI feminist perspective, as Winkle et al. point out, means being sensitive to power structures [18]; this can be at an institutional level (ie systems and institutions) and individual level (based on gender, sex, (dis)ability). Power is therefore referred to as the *current* configuration of structures, whereby whilst it might benefit some groups, it might also oppress others. The 'HRI' part is also important here, as robots through their embodiment, bring their own challenges. Consequently we, as researchers and designers, need to consider the context in which the robot is used in as well as how people (for eg researchers in our study) position robots in a setting. Overall, we summarise our theoretical lens, a motivating start-point but also the lens through which we analyse and discuss our results, as Social Justice-Oriented Design and Feminist HRI. This uses the concepts of *recognition* of stakeholders and practices as well as *accountability* of HRI research for health care, within *power* structures which will position robots in certain settings and impact various stakeholders in health care.

5 Method

5.1 Research Questions

The overall research question we explore can be summarised as: how do HRI researchers position informal caregivers within studies relating to HRI in/for health care?

We pose four sub-research questions that help us answer the main question:

- 1. How are researchers motivating studies in health care when including different stakeholders?
- 2. Which stakeholders are included in HRI studies for/in health care?
- 3. What methods are HRI researchers using when including participants for/in health care?
- 4. How and at what point are caregivers included in health care HRI studies?

5.2 Choice of Literature Corpus

Influenced by previous studies that have done so (for example, [13-16, 54]), this study is limited to the ACM/IEEE HRI conference. The nature of the conference is to showcase the latest pioneering research within HRI, operationalised via e.g. limited acceptance rate, therefore we think it is reasonable to assume that this research should be a sample of community-assessed good-practices within HRI in/for health care. Pragmatically, the nature of our explorative qualitative analysis, namely thematic analysis (explained in subsection 5.4), requires significant time and resources, resulting in setting some upper bounds on the amount of literature we can include. As can be seen below, the terms used to collect the literature were very broad, leading to a relatively large number of hits which required manual assessment. This was intentional, in order to 'catch' papers that might have somehow included informal caregivers without mentioning them as such, and to demonstrate which stakeholders are key in health care HRI research. Finally, we note that there is not one clear obvious alternative venue or venues that might be more or less relevant. For example, the CARESSES project published in the International Journal of Social Robotics but do not once mention informal caregivers [55], however they also published in Archives of Public Health where they explicitly mention how they recruited informal caregivers [44]. Consequently, we assumed that using our sample following the PRISMA method (explained below) within the ACM/IEEE HRI conference would lay groundwork on the relevant stakeholders and focusing especially on how the HRI community is accounting for informal caregivers within health care.

5.3 Delimiting the Search and Screening

To complete this systematic literature, we relied on the PRISMA method (see Figure 1), whereby based on keywords, we identified, scanned and included articles according to our inclusion and exclusion criteria [56]. Using the ACM Digital Library, we searched for research articles only within the HRI conference proceedings to find any abstracts containing the following keywords: ("health care" OR "healthcare" OR "assistive" OR "impairment") AND ("care" OR "assist*"). Each keyword choice was decided between authors, which includes engineers and scientists, all of whom are critical scholars.

We used boolean operators in order to group terms, those include the use of "OR" in the same bracket to have similar terms searched for, and in between brackets we used "AND" to combine layers. The first set of brackets, ("health care" OR "healthcare" OR "assistive" OR "impairment"), were decided to saturate the field of health care, where we combined different terms that are similar to care within HRI literature; The "AND": due to the amount of results from this first layer, we used the "AND" operator to bring an additional layer; The second set of brackets: ("care" OR "assist*"): this bracket was specifically to ensure that the robot was deemed to either care or assist in some way. The asterisk (*) was used in order to also get results which used the stem word 'assist', including terms such as "assistance" and "assist".

The results were deemed broad enough, with a total of 187 hits. All the literature from the results were then exported onto EndNote 20 to be able to screen the documents easily. Figure 1 shows the screening process, which totalled to 52 studies to be reviewed and analysed.

We had clear inclusion/exclusion criteria. The inclusion criteria consisted of academic literature which were within our health care definitions and referred to care robots. More specifically, the study was about bringing a robot into a certain care setting; care was defined as health care, meaning that the study either looked at improving current care practices, improving quality of care, bring some independence to the user or loneliness and companionship, supporting or motivating the user or/and reduce workload within healthcare settings. Furthermore, studies must have used a robot, this can mean anything from a robot which aims to assist with exercises and rehabilitation to a robot that monitors the care receiver, or aid the caregiver. The use of the robot in the study can be a hypothetical robot; videos/images of robots; and an actual physical robot. Finally, the study must have participants.

The exclusion criteria were studies not within health care, for example the study contained our keywords in a different context but in educational settings; literature reviews; studies which evaluate the use of robots but do not consider technical aspects to do (eg ethnographies); and abstracts.

5.4 Reviewing and Analysing the Literature using Thematic Analysis

To achieve our aim, we used a thematic analysis approach to review and analyse our collected literature, in accordance with Braun and Clarke [57]'s five steps (Familiarisation, Initial Coding, Theme Search, Theme Review, Naming and Definition). The first step was familiarising ourselves with the data. This allowed us to create initial



Fig. 1 PRISMA Flowchart Diagram of Selected Studies [56]

themes and create a codebook for the second step. A codebook defines codes as well as delimits them, and thus creates a deductive approach to coding. As the third step (theme search), we transferred the articles onto NVivo, a software used to qualitatively code data (herein academic articles) and used the codebook to code themes within the articles. This was an iterative process to ensure our codebook reflected the literature reviewed (theme review and naming and definition).

The use of NVivo yielded qualitative and quantitative data that was easily reviewable. This allowed for one author (LT) to review and code all articles following iterative development of the codebook with a supervising author (KW). Together, LT

(sociolegal scholar) and KW (HRI scholar) bring a combination of critical and HRI perspectives. The codebook was continuously re-iterated between these two authors until saturation was reached for the purpose of our literature review.

Since the data was coded on NVivo, it was possible to go beyond the binary and add the papers into various categories within the same overall code (who was investigated in the research: care receivers, formal caregivers and other experts for example). All the results from the coding were quantified in an Excel Spreadsheet after having done all of the qualitative coding, which is available in the additional materials.

6 Results

In our results, we focus on the four following themes – representing each subsection – that shed light on how stakeholders are positioned within HRI studies, with a specific focus on informal caregivers:

- 1. *How works were justified* within the opening paragraph, i.e. how authors motivated bringing a robot into the care context,
- 2. Who (in terms of role within care) was specifically investigated within the study,
- 3. *How* studies were conducted and if/how this varied when investigating different stakeholders,
- 4. To what extent (in)formal caregivers are considered in the reviewed articles overall.

6.1 Overview of Reviewed Articles

This subsection sets the scene on the article that were reviewed, and how the articles themselves set the scene for their study.

6.1.1 Background of Reviewed Articles

The articles within the health care domain started getting published in 2008 within the HRI Conference, as can be seen in Figure 2, which is two years after the beginning of the HRI Conference. Out of the 52 articles reviewed, eight studies included informal caregiver (totalling to 15% of the time) and 20 studies included formal caregiver (totalling to 38% of the time).

The first appearance of formal caregivers were in 2010 – which is also the only publication on health care that year. After that, they are sporadically included as participants, and from 2017 formal caregivers have been constantly included in health care studies. The first publication to include informal caregivers was in 2014. However, the next time informal caregivers are investigated is in 2020, six years later, where there is also seemingly an exponential growth of HRI studies in health care. Overall, informal caregivers are included at a much lower rate in comparison the amount of HRI publications in health care.

Furthermore, articles either looked at care for: older people, over 65s (35%), adults (12%), children (13%), all age groups (42%), and staff (12%). For studies that investigated informal caregivers, the figures are the following: 35% for older people, 0% for adults, 25% for children, 25% for all age groups, and 0% for staff. Whilst for studies



Fig. 2 Graph showing the number of publications between 2008 and 2023

that investigated formal caregivers: 50% for older people, 15% for adult, 5% for children, 15% for all age groups, and 25% for staff (representing all the articles which look at how to support staff, except for one which also aims to support formal staff but does not investigate formal caregivers [58]). All informal caregivers investigated were over the age of 18.

6.1.2 Motivation behind Reviewed Articles

Motivation as a theme is used here to pinpoint how authors motivate their research within their opening paragraphs. We identified *alarming narratives*, lived experiences, and further (ing) current robotic research. Alarming narratives refers to the use of statistic (such as the change in demographic for example), or the use of "doom and gloom" to demonstrate that there is a need for robots to enter that domain. An example of this: "According to a recent report from the World Health Organisation (WHO), there is currently a global shortage of 7.2 million healthcare workers, and this number is expected to grow to 12.9 million by 2035" [59, p.243]. Lived experience, refers to helping the reader understand either the technology or the context from a care role perspective, without necessarily using academic referencing nor being specific to HRI. A good example of this is by Nanavati et al. [60, p.24]: "Take a moment to recall the last time you shared a meal. What made it meaningful? The company, the food, the ambiance? The stories that were told, relationships that were strengthened, milestones that were celebrated? If you were asked, 'How does it feel to eat socially?', you might say it is a pleasant experience". The final category showcased here is further current robotic research, which is where the justification refers to previous (HRI or related)

articles looking at a similar issue within the domain, positing the current work as developing on/responding to those previous works.



Fig. 3 Chart comparing motivation(s) relied on by the studies to set the scene

Figure 3 illustrates the three categories spoken above. Alarming narrative is used by most articles, nearing 50% of the time (n= 25 of all studies rely on this), although it is used more often when the studies include caregivers than care receivers, especially when formal caregivers are participants (60% of the time). With regards to studies involving informal caregivers as participants, it is used 50% of the time ([61-64]). The use of alarming narrative for studies including informal caregivers represents some recognition of their importance in healthcare, as Lee et al. [64, p.475] show in their opening paragraph: "10% of older adults in the US live with dementia. Dementia requires substantial amounts of care; often, this is largely informal care provided by the patient's family members. About 85% of the dementia patients reside in their own homes rather than in nursing facilities. Unpaid family members are the most common carers for persons with dementia; this continues even after persons with dementia move to residential care settings". Lived experience is also used relatively frequently (38% of the time -n = three out of eight studies) specifically by studies including informal caregivers. In contrast, lived experience is referred to in around only 15% (n = eight out of 52 studies) of the overall articles reviewed.

Finally, motivations based on *furthering current robotic research*, were not seen so often, at least within the opening paragraphs. Approximately 17% of all studies (n= nine out of 52 studies) included some such motivation, **none of which** included studies relating to informal caregivers. This already perhaps points to a lack of previous work specifically considering informal caregivers in HRI research.

6.2 Who is included



Fig. 4 Graph showing and comparing how many groups are included

Across all articles reviewed, 71% of studies included care receivers as participants (n=37 studies out of 52 include care receivers as participants). Formal caregivers were included in 38% of articles (n=20 out of 52 studies), and informal caregiver were investigated in 15% of articles (n= nine out of 52).

6.2.1 Studies Investigating One Stakeholder

Different groups are mapped out in this part to showcase who is investigated in the study itself, which is illustrated by Figure 4. Over half of all articles reviewed (54%) solely investigated one stakeholder group, with care receivers being the most common [36, 58, 60, 65–89]. Specifically, out of the 54% investigating a single group of participants (equating to 28 studies out of 52). Out of those 28 studies, 61% investigated care receivers, 21% investigated control groups, and 18% investigated formal caregivers When looking solely at the articles that investigate formal caregivers, those studies will investigate formal caregivers as a sole group 25% of the time (n= studies out of 20 studies). In contrast, **no studies** investigated informal caregivers only.

6.2.2 Studies Investigating Two or More Stakeholders

33% of the articles reviewed investigate two groups (equating to 17 studies out of 52), and here the dynamic slightly changes [62, 64, 90–104]. 55% of all studies investigating formal caregivers also studied another group of participants [90–99, 104]. Similarly,

50% of studies which included informal caregivers also included one other group of participants [62, 64, 101, 104].

Finally, 14% of studies investigated three groups or more, equating to seven studies out of the 52 studies. Interestingly, only one such study does not include caregivers [105], all others do. Regarding formal caregivers specifically, 20% of studies (n= four out of 20 studies) additionally investigate two or more other groups of participants [59, 61, 106]; and four groups [107]. Turning to informal caregiver, this number increases substantially to 50% (n= four out of eight studies), as this group tends to be investigated in addition to two or more groups of stakeholders within studies (included in three groups: [61, 63, 108]; included in four groups: [107]).

6.3 Methods Used

6.3.1 Types of Methods Used

Methods here refers to approaches HRI researchers relied on to answer their research question(s). We show results for studies which did their data collection (1) in the wild, (2) through interviews or focus groups, (3) in labs or lab-like settings; and (4) through workshops. We group interviews and focus groups together on the basis that both typically target relatively homogeneous participant groups (e.g. "therapists"), unlike workshops – hence they are separated. As an example, focus groups were held in two different eldercare institutions with professional caregivers to inform robot design within that setting [82]. In contrast, an example of workshops is one where children and their parents were brought together by HRI researchers to co-design socially assistive robots to mitigate distress and pain in pediatric care and understand their perceptions [62]. Figure 5 showcases the four methods and how they are used in comparison to all studies, studies including informal caregivers and studies including formal caregivers.

Looking at studies conducted in the wild, this method was used 19% out of all the articles reviewed (n= 10 out of 52 articles). Six out of those ten studies (60%) included formal caregivers in their studies; five of the 10 studies (50%) are conducted in elderly care (50% of studies conducted in wild); one study (10%) is conducted in hospitality settings [95]. None of the studies conducted in elderly care facilities and hospital settings include informal caregivers. Only two studies (20% of studies conducted in the wild), are in a setting where the robot is left to use in a domestic setting for some time [76, 101], of which one study is for older people. Informal caregivers were only directly investigated in one of the in the wild studies [101], whilst for studies investigating formal caregivers, in the wild was a method used in 30% of the time.

Regarding interviews and focus groups, these methods were used in 42% of all articles reviewed (n= 22 out of 52 studies). It is the most popular method again when formal caregivers are being investigated, used 70% of the time within articles studying formal caregivers (n= 14 out of 20). However, it is only used 25% of the time when informal caregivers are being investigated (n= two out of eight). Methods using lab/lab-like are used over 52% of the time in all articles reviewed. These methods are not as popular for studies investigating formal caregivers (used 25% of the time), but used more when informal caregivers are investigated (38% of the time, n= three out of eight studies). Finally, workshops are the least typical method for most studies, as



Fig. 5 Chart comparing various methods relied on depending on who is included

it is only used 12% of the time in all reviewed articles. In articles investigating formal caregivers, it is used 20% of the time. However, it is the most popular format for studies investigating informal caregivers, used 50% of the time.

6.3.2 Number of Methods Used

It is worth noting that 58% of all studies use at least two different methods (n= 30 out of 52 articles), thus cross-pollinate. For example, no study conducted in the wild, or studies using surveys solely relied on that method. Studies relying on one method mostly tend to use interviews (45% out of all the studies using one method). Looking at trends with formal caregivers, 60% of studies use at least two methods (n= 12 out of 20 articles). With informal caregivers 62,5% of the time at least two methods are used (n= five out of eight). Turning to formal caregivers, they are involved in most studies which use three methods – out of the 10% of all articles using three methods or more, 75% of them look only at care receivers and formal caregivers. Two of those reflect on long term deployment of robots, either for older adults with dementia [98], or for rehabilitation for post-stroke patients [95] and one, which is outside of the medical institution, to bring well-being robots into the workplace [92].

6.4 Caregivers' Involvement in Studies

Here we want to zoom in directly on caregivers, demonstrating also how we have defined informal caregivers in studies –even if researchers did not call them as such–whilst also showing a comparison between how much attention has thus far been paid to formal caregivers and informal caregivers.



Fig. 6 Graph showing how many different methods are used

6.4.1 Identifying Informal Caregivers

As a reminder, we define informal caregivers as "A person of any age, adult or child, who provides unpaid support to a partner, child, relative, or friend who couldn't manage to live independently or whose health or wellbeing would deteriorate without this help. This could be due to frailty, disability or serious health condition, mental ill health or substance misuse" [9, p.1]. Put differently, a member of the family or a friend can be viewed as an informal caregivers if the support that they bring allows the care receiver better overall well-being and health. We do not include parenting as a general rule, however, parent-carers are an important – often overlooked – group, denoted by situations where the child would not be able to manage their condition without the help of their parent. These definitions are important to highlight, since out of the eight studies which include informal caregivers directly in their study, summed up in table 1, three of them (38%) do not actually refer to them as informal caregivers. Therefore it seems fitting to explain why we categorised them as such.

Out of the five studies that identified and explicitly named informal caregivers [61, 64, 104, 107, 108], three of them work with a robot application targeting users of all ages, whilst the other two consider elderly care specifically; furthermore, two use alarming narrative to situate their study [61, 64] and two others use lived experiences [107, 108]. Regarding the three which did not identify informal caregivers as such, two of them consider a robot application for children and one for all ages. Only one study used lived-experiences to situate the study [101]. This study, by Tsoi et al. [101], deployed robots to children during the Covid-19 pandemic, and opens by speaking of the social isolation that occurred due to restrictions imposed to limit social contact.

Title of Publication	Year	How informal caregivers are
		included
A mixed-method approach to evoke cre-	2014	Scenario-focused workshops with differ-
ative and holistic thinking about robots		ent stakeholders, which included informal
in a home environment [107]		caregivers.
Robots for joy, robots for sorrow: com-	2020	Co-designed robots through workshops
munity based robot design for dementia		on caring for people living with dementia,
caregivers [104]		which included informal caregivers and
		geriatric nursing students.
Transfer depends on acquisition: ana-	2020	Study included people who had experi-
lyzing manipulation strategies for		ence feeding other people to test out the
robotic feeding [63]		technology.
Co-designing Socially Assistive Side-	2021	Informal caregivers were included in
kicks for Motion-based AAC $[108]$		workshops to co-design assistive conver-
		sation technology.
Challenges Deploying Robots During a	2021	Toy robot was brought into homes to help
Pandemic: An Effort to Fight Social		with child isolation during covid, parents
Isolation Among Children [101]		provided information needed.
Understanding Design Preferences for	2022	Informal caregivers co-designed to under-
Robots for Pain Management: A Co-		stand features both child and parent
Design Study [62]		would want for pain management robot
		application.
Impacts of Robot Learning on User	2023	Compares perceptions from general pop-
Attitude and Behavior [61]		ulation to caregivers of robot applications
		in the home for health care purposes.
Reimagining Robots for Dementia:	2023	Explores the varying views of persons
From Robots for Care-receivers/giver		living with dementia and their informal
to Robots for Carepartners [64]		caregivers and the importance of taking
		both into account for the robot design.

Table 1 The role of informal caregivers in studies that do include them

The reason we categorised Tsoi et al. [101]'s study as directly including informal caregivers is because the parents were the ones that assessed if their child felt lonely and how this could be improved, whilst also being the ones to set up the device, and fill out various surveys. The unusual circumstance, namely the pandemic, is what made their role go from being a parent to also taking on a more caregiving role as they became the ones that had to create the adjustments needed for children to live a healthy everyday life away from peers, activities and school, to name a few.

The other two studies relied on alarming narrative to set the scene [62, 63]. One of the studies, by Gallenberger et al. [63] investigated feeding strategies for robotic feeding. To achieve this, they recruited 25 people, of which eight, the paper specifically points out, had experience feeding others. It is somewhat unclear if the participants were informal caregivers, however, the fact that the researchers pointed it out demonstrates that the researchers suspected that the participants relied on their previous

experience to understand if this robotic tool would be useful. There is no clear way for us to know what capacity these people were feeding others, but as it is usually specified in other studies if the participants have formal qualifications which was not the case here, we therefore decided to categorise them as informal caregivers. Another study, by Zhang et al. [62] undertook a co-design study which includes parents and their child to understand design preferences for robots for pain management. This study directly looks at parents to help understand how to manage pain (for example, for surgical recovery) for children. The parents are not identified as informal caregivers, however, since this study is to ensure a safe recovery from pain for children to be able to avoid long-term consequences due to the initial pain in childhood, it shows that parents in this context play a role that falls more in line with parent-carer, and thus, that of an informal caregiver.

The definition of informal caregivers also allowed us to reflect on how robots were sometimes defined in studies, which aligned with the role of informal caregivers. For e.g.: "Robotic systems may assist older adults in maintaining their independence, reducing healthcare needs, providing everyday assistance, and promoting social interaction" [72, p.19], or "Furthermore, they [children with diabetes] greatly enjoyed working with the robot and came to see it as a helpful and supportive friend" [66, p.463], or "[subsection: Intended Roles: Friend/Partner]: Robots can accompany older adults during PA [physical activity] (e.g., walking, creative hobbies) to counter loneliness and augment the PA as a social activity that may not be otherwise possible due to an inability to keep up with a partner or group in a meaningful way." [106, p.510].

6.4.2 Identifying Formal Caregivers

In stark contrast, identifying formal caregivers demanded a lot less reflection on whether we should define that group as formal caregiver or not. Indeed, that group was named according to their profession, this allowed us to categorise and differentiate between "formal caregivers" and "other experts". For example, [109] included formal caregivers as they interviewed medical professionals that practiced medicine within the medical institution, as well as other experts, gender scholars who looked at and reflected on medical practices. Whereas [36] hosted focus groups and interviews with physiotherapists, occupational therapists, speech therapists and sports rehabilitation therapists, all of which fall under formal caregivers as they practiced within the healthcare institution and required certification.

With this understanding, we can now spotlight different levels of involvement between caregivers.

6.4.3 Level of Involvement of Caregivers in Studies

Up until this section 6.4, we have solely mentioned and reflected on stakeholders that are expressly investigated in the study. However, there are also interesting nuances on how the caregivers are involved and identified, we therefore used four different categories: "no mention of caregiver"; "acknowledgement of caregiver"; "low level of involvement"; and "high level of involvement", to show at what level caregivers are mentioned (as depicted in Figure 7).



Fig. 7 Different levels of involvement of various stakeholders within HRI studies

No mention of Caregivers

10% of articles do not mention formal caregivers at all (n= 5 out of 52), whilst 29% of articles (n= 15 out of 52) – nearly one third of papers collected – do not mention informal caregivers.

Since there is such a high proportion of studies that do not mention informal caregivers, we quantify those that exclude them fully from their study. Four studies (27%) are situated in elderly care [71, 78, 82, 89]; one application is situated for children [67]; eight studies are targeted for all ages [65, 69, 70, 73, 84, 96, 100, 102]; and, three (20%) looking at robotic application for staff [82, 89, 99] — of which two are directly for medical staff. Turning to groups involved, 10 studies out of the 15 studies look at one group of participants. Within those 10 studies, seven studies look at care receivers; two studies look at formal caregivers; and one at control groups. The other five studies include two groups, two of which look at formal caregivers and care receiver; two studies investigate care receivers and control groups; and, one investigating formal caregivers and control groups.

Acknowledgement of Caregivers

Acknowledgement of caregivers is quite high for both formal caregivers (29%, n=15 studies) and informal caregivers (46%, n=24 studies). This means that the caregivers are not involved in the studies, however they are mentioned in some way. With regards to formal caregivers, this is done either through responses within interviews which name them [36, 60, 62, 105], or researchers mention that they hope the robot will help healthcare professionals [88], or the researchers want to imitate the medical setting, ie



Fig. 8 A comparison between caregivers regarding how either (1) involved they are in the study, (2) acknowledged, and (3) not mentioned in the collected studies

a therapy session or physio [63, 65, 73, 85, 87, 102], or they are identified in previous relevant work [72].

There were studies that were unclear about whether they targeted informal caregivers or formal caregivers, but acknowledged that it would help support caregiving responsibilities [74, 79, 86]. We therefore categorised them as both types of caregivers in our analysis.

Within the 24 studies that do acknowledge informal caregivers within our definition, 15 studies (62,5%) refer to them in terms of their relation to the care receiver (eg 'family' or 'friend') rather than as informal caregivers. Within those studies, some take for granted how they would create informal caregivers, or create more workload to informal caregivers. An example is with regards to setting up the study: "[section "inclusion criteria]: Background: At least one friend/family contact who is willing to connect by video-call; time to spend on GenieConnect, not too busy; has good WiFi coverage" [76, p.1096]. Another example is to play a rehabilitative game "Another possibility would be to have one patient use a robot while another patient (or a therapist, or even a healthy friend of the patient) uses a simpler haptic interface" [85, p.453].

Furthermore, regarding informal caregivers, their acknowledgment is summarised as: families helping set realistic goals to support the use of social robots for the care receiver [91, 93] or plays along [85] or can view and monitor progress [36]; family members could help set up the system [76, 107]; feedback from parents at the site the research is conducted – that is further elaborated on in a different paper [88]; recognising families as tertiary users of robots for its application in pediatric hospital setting [59]; needing to also account for robot interactions with family members [81]; recognising that potential caregivers are not accounted for [90]; responses from interviews

with care receivers [97, 105] or interview questions to care receivers include level of support they get from friends and family [77]; speaking of caregivers on a general level [80, 87, 98, 103]; referring to past literature which acknowledges families providing care [58, 72, 83, 95]. Finally, one study specifically speaks of how they tried recruiting informal caregivers, but were unable to do so [106].

Low Level of Involvement

Turning to low level of involvement, the difference between low and high, is that the caregiver is not relied on but has a more passive role to help with the running of the study. This low involvement occurred 15% of the time for formal caregiver (n= eight out of 52 articles). Whilst for informal caregivers, they are involved at a low level 10% of the time (five out of 52 articles).

Within low involvement, we included practices within the healthcare institutions which would require healthcare expertise in order to design it into the robot, as this demonstrates how expertise transpires into studies even if those experts are not directly participants. One study does so by automating Confusion Assessment Method for Intensive Care Units (CAM-ICU) used to detect delirium into a robotic system [83], whilst another 'induced' anxiety, through the Trier Social Stress Test for children, to demonstrate that robots can help improve children's mental health [103]. Low involvement also included health care workers that were not directly involved in the study, but were at the heart of the study in order to ensure that the tool could be used for them, an example is a study investigating how to use teleoperated robots in highly infectious hospital environments in order to allow health care workers to perform some of their duties at a safe distance [58]. A further two studies asked formal caregivers to propose participants who could take part in the study [66, 76].

We flagged a couple of studies that did not make it clear if the involvement was about informal or formal caregivers. One such study asked participants to plan a surprise party for their caregiver, and the participants were asked to think of who this caregiver should be and that chosen caregiver would be allowed to monitor the participant through the technology the participant had been assigned in this experiment [68]. Another study mentioned that caregivers accompanied the care receiver, herein the participant, in order to (1) go through the consent form, (2) go through the study protocol, and (3) attend the trial in order to engage in normal conversation with the participant [75].

Within studies that somewhat involved informal caregivers, two out of the five studies have been explained above. The three other studies, one study directly points out the involvement of the parent's participant in order to sometimes help clarify what the participant said during the interview [60]. Another speaks of an introductory meeting with the primary caregivers and family members of the prospect participants ahead of the study [94] (note that formal caregivers were part of this study, but not the potential informal caregivers). Finally, one study which investigated the use of a social robot to motivate and support children with diabetes keeping a diary, they speak of the important role parents play but also that some parents supervised the children whilst they were writing up their diary entry for the study [66].

High Level of Involvement

High level of involvement is the highest category that can be achieved in our classification, as it means that the caregiver is either (i) a participant or (ii) heavily relied on to directly mould the study in some shape or form. The analysis of all articles showed that although 38% directly investigate formal caregivers, four more studies directly rely on formal caregivers to ensure that the study is successful, meaning that 46%of studies overall highly include formal caregivers (equating to 24 studies out of 52 articles). Out of the studies that do not include formal caregivers as participants, one speaks of direct collaboration between roboticists and occupational therapists to create a flexible interface for a wheelchair mounted robotic arm [70], another was relying on a formal caregiver to interview some of the study participants in order to compare to a robot that asked the same questions [77], another study relied on caregivers in the facilities for children with autism to help design tasks for the robot [67], and finally one study relied on the executive director, who had also been a social worker, who directly helped with the question formulation, what questions should be included as well as collecting feedback from participants [64]. In contrast, no other informal caregivers were highly included in the studies, except for the ones that have already been mentioned (15%, equating to eight out of 52 articles).

This means over 60% of all studies collected, formal caregivers play a vital role in health care HRI either through high level involvement or low level involvement, whilst for the same level of involvement that number is 25% for informal caregivers.

7 Discussion

7.1 Applying the Theoretical Lens to the Findings

Our theoretical lens, deriving from Social Justice-Oriented Design and Feminist HRI, allows to map out different characteristics for stakeholder(s), context, and/or institution(s). This framework has enabled us to recognise various stakeholders, highlighting informal caregivers in particular, within different health care contexts; whilst also adding notions of accountability and being attentive to power structures. As can be seen from the results, our analysis dissected 52 research papers, where we set out to understand how the researchers set the scene, who they involved, what method they used, and how/if caregivers are mentioned. We were also candid about the complexity of categorising caregivers, which at times lead to subjective categorisations. Overall, this was an exercise of disentanglement in order to answer our overall research question, *How do HRI researchers position informal caregivers within studies relating to HRI in/for health care?*, and show a complex and nuanced understanding of informal caregivers within HRI.

The Who in HRI Health Care Studies

Our results show a trend in HRI work focusing on health care, with the majority of articles (71% of all articles reviewed) considering the care receivers' needs and perspectives. This is of course a positive trend, and should not be viewed as a critique. However, the results point at the complexity of who else is involved and how, especially

when looking at informal caregivers. Overall, informal caregivers are on the periphery of HRI health care research. If comparing to formal caregivers, formal caregivers are investigated 38% of the time, as opposed to 15% of the time for informal caregivers. Whilst 75% of the time, informal caregivers are either acknowledged in studies or not mentioned, in comparison to 40% of formal caregivers are either acknowledged or not mentioned. This is a significant difference, especially when considering the vital role informal caregivers play as co-providers of care [1].

Setting the Scene in HRI Health Care Studies

Mapping the opening paragraphs of the reviewed articles demonstrates how the studies motivated their study to grasp people's attention. Out of all the studies which investigated informal caregivers, none relied on previous robotics research that they wanted to further – a practice that was somewhat used in other articles reviewed. Instead alarming narratives and lived experiences were used. It is worth emphasising that the use of lived experiences in studies investigating informal caregivers was a lot more than the average, see Figure 3. The use of alarming narrative has already been observed by other researchers, such as Maibaum et al. [40], Wright [42] and Pedersen, Reid and Aspevig [41], as it allows roboticists to show that there is a need for the robots to enter the field and find solutions. However, the use of lived experiences is not something that has previously been captured. Typically, the use of this type of practice is part of feminist approaches (see [110]), as it allows to show more individual perspectives. Put differently, lived experiences allows to show that there are other perspectives that could (and probably should) be taken into account, which could be explained by the fact that informal caregivers lie outside healthcare institutions but are still part of it from the sidelines. However, if this practice is relied on more than usually done in HRI studies, this could diminish the importance of informal caregivers to the HRI community.

Methods Used for answering HRI in/for Health Care

In relation to methods, we demonstrate how the studies were being conducted when certain stakeholders were involved. With regards to informal caregivers, the most popular methods used were either (1) lab or lab-like studies, or (2) workshops. In contrast to this, studies investigating formal caregivers were mostly using interviews or focus groups, closely followed by in the wild studies, see Figure 5.

The findings on the methods used, although a lot cross-pollinated, still show important trends. For example, when studies were conducted in the wild, 60% of such studies included formal caregivers. Within those, the studies tended to ask formal caregivers their reflection on the use of robots. Harrison and Johnson [111] point that in the wild studies allow levels of variation participant responses, with regards for example, to robot encounters they enjoyed and did not. However, informal caregivers – but not named as such – were part of in the wild studies once. Instead, it was much more likely that informal caregivers were involved in lab or lab-like studies and workshops (workshops being the most popular approach to investigate informal caregivers). Though, Harrison and Johnson [111] point out, lab studies tend to recruit participants who are more positive towards robots; and we add that this is likely to extend to workshops.

How were Caregivers Included in HRI Health Care Studies

Regarding who was investigated, we showed how many groups per study were investigated. When looking at solely one group of stakeholders being investigated in HRI health care studies, it was found that 21% of the time it was care receivers; 18% of the time it was formal caregivers; but informal caregivers were never investigated as one group. Instead, informal caregivers were part of studies which included two groups or more; meaning that they were part of studies which investigated more stakeholders than the average, see Figure 4.

Moreover, when informal caregivers were included, see Table 1, some of them had quite passive roles. This might not be so surprising, especially as literature reviews (including critical ones: [40-42]) do not tend to put much emphasis on informal caregivers' role within the HRI-health care eco-system. However, within health care research generally, the role of informal caregiver is unquestioningly necessary to the sustainability of healthcare institutions [1, 3, 10, 11], with many studies solely looking at them, for example: [2, 3, 10, 11, 21-23, 25].

7.2 The Complex Role of Informal Caregivers: Vital but Under-Recognised

Informal caregivers are part of the HRI health care eco-system. By showcasing level of involvement of caregivers in our results we pinpoint how informal caregivers play a role, even if passive. Regarding 'low level involvement', we pinpoint that they help from a recruitment perspective, or allowing the care receiver (participant) to take part in the study, or supporting the research process generally. This low level involvement occurred 10% of the time for informal caregivers in all of the articles, which is similar to formal caregivers (15%) of the time). This reliance also shows that there may be an expectation on caregivers to be around, without necessarily being investigated. This is further evidenced when the informal caregivers are acknowledged in studies: they are acknowledged 46% of the time in all the studies collected, and within those times, they are sometimes either expected to be available for the study or they are expected to help with the set up of the robotic software once it is available. Although this is also the case some of the time for formal caregivers, formal caregivers will be monetarily rewarded for their time spent, since it is/will be part of their job role. Furthermore, studies conducted in the wild within institutional settings, have not included informal caregivers. Yet, when formal caregivers are included in these studies, it is to reflect on their experience of this robot on their everyday work life. This demonstrates that robotic applications for formal caregivers are being much more proactive in getting their view alongisde the creation of the robot application; yet, methods used when informal caregivers (workshop being the most popular) were more about trying to understand how to account for them at design stage. This is also echoed by the number of groups involved, where informal caregivers are never investigated as a sole group, unlike care receivers and formal caregivers.

Relating this to previous work, informal caregivers are viewed as co-providers of care [1], as they will be the ones doing the bulk of the care outside of institutional care,

and still provide support within institutional settings [23]. This insinuates that, actually, HRI studies should and need to account for informal caregivers when they are also looking at formal caregivers. This is key as HRI Feminism and Social Justice-Oriented design point out, that there is a whole eco-system behind every HRI research, and it becomes the HRI community's task to ensure a more diverse pool of participants is being included and to engage in the political structure of our research [16-18]. Furthermore, as it stands, the HRI community could potentially further burden informal caregivers, with issues such as information overload [25], difficulty accessing services, and at times care receivers, due to a global pandemic [24] – or in reverse, more care responsibilities are on the informal caregiver, as the care receiver cannot access professional services. In line with Winkle et al. [18] and Dombrowski, Harmon and Fox [17], we add that the HRI community should attempt to align with societal challenges beyond that of statistical narratives which tend to overlook some voices. Said differently, although there might be an ageing population and a decrease in healthcare professionals available, this is likely to create – as well as put more pressure on – informal caregivers and we have a duty to account for it.

7.3 Recommendations: Including Informal Caregivers to Better HRI in Care

The findings section attempted to demonstrate an exhaustive set of results, which we rely on to be able to give high-level recommendations on informal caregivers within HRI in health care, and maybe beyond.

Firstly, it is key to involve informal caregivers more in studies, either through high level involvement or low level. High level involvement would mean either recruiting informal caregivers as participants, or directly collaborating with them in some way to help steer the study in health care. We do understand that informal caregivers are difficult to identify and/or recruit. Accordingly, we suggest for low level involvement, researchers could disclose if they have any experience as informal caregivers, and say how this influenced their choice on how to account for informal caregivers in their robot application. This would be in similar style to low level involvement of formal caregivers, where researchers used their expertise as formal caregivers to steer their own health care research (for example, inducing anxiety in children). Although this has not been done (yet) by researchers looking at informal caregivers, it is likely that some of us have experience with such a role, and using that expertise explicitly in research studies could be a way to involve more openly informal caregivers.

Secondly, as researchers, we need to acknowledge the impact our potential use-cases will have on informal caregivers. Clearly, informal caregivers play a significant role in the health care eco-system, and will continue to do so in HRI health care applications. As pointed out in the results, we expect them to be able to help either (1) set up the device, and/or (2) set goals, and/or (3) play a game with the care receiver, and/or (4) give feedback. And likely, in the future, repair the robot – such as re-boot the robot – if needed. However, they will be expected to do so without necessarily being monetarily compensated for their time and effort. This is in contrast to formal caregivers, who are, and will be, employed to be trained on how to use the robot as well as paid to do it. Therefore we need to be aware of this as a community and explicitly acknowledge

the potential added – or new – workload onto informal caregivers, and how we can do so in a mindful way for all stakeholders involved, which has to include informal caregivers.

Thirdly, we recommend ensuring that bringing tech solutions to care receivers, which will likely impact friends and families, need to be recognised in policies and guidelines. The definition of caregivers offered in this paper did not refer to technology in any way, yet, as pointed out in the paragraph above, it will increasingly be required that informal caregivers help with the set up and use of tech solutions. Although we do not map out exactly how the robots were used for care receivers, the way both formal and informal caregivers were framed in our findings demonstrates that they will play a pivotal role in the deployment of robotic solutions. Whilst formal caregivers are being looked at (for eg [112]), the same needs to be done for informal caregivers to advise on necessary policies and guidelines recognising informal caregivers in this realm as well as ensure that this technology is suited to their needs.

Fourthly, from the dataset collected to do this research as well as from the relevant literature in this field, informal caregivers are seemingly an overlooked group in HRI and thus present a research gap. As the HRI community, we can leverage on our own power – that is who we decide to include and how. This will likely require to start thinking about including informal caregivers from the moment that the research in health care is being designed, in order to ensure that they are included in ethics applications and the recruitment process. Regarding recruitment, one possible way of doing so would be to ask care receivers to also ask their caregiver to take part in the study with them. And/or conduct the study in a way that would be least demanding on informal caregivers, for example, ringing informal caregivers to ask for their feedback on a certain application rather than asking them to fill out a survey or come to a location on a certain time and date. The latter would be well suited for in the wild studies, where informal caregivers might have interacted with the robot at various points during their visits. In turn this would allow for HRI research to respond to more bottom-up demands, rather than institutional ones [40, 46].

Finally, echoing the points above, there is a potential to use more technology in real life context for informal caregivers themselves. As more HCI applications try to cater specifically for informal caregivers [28–30], this shows the possibility for HRI to also look at application-specific for informal caregivers. HCI justifies this research as informal caregivers use technology in any case to help with their care role, thus there is a possible demand for this [27]. In alignment with this, research has demonstrated that if the technology is not well adapted then care receivers and informal caregivers will steer away from it [28]. This leaves the possibility for HRI to find ways to support informal caregivers' well-being, and in turn improve quality of care for care receivers.

8 Conclusion

This study has aimed to disentangle HRI researchers' practices within health care through a systematic literature review, using a qualitative analysis method, on 52 health care HRI studies in the ACM/IEEE International Conference on Human-Robot Interaction. The process of disentanglement lead to dissecting research papers to showcase a narrative on four different facets: 1) how the researchers capture people's attention to demonstrate the need for this application as well as for 2) who the robot application is targeted for and also mapping out how many groups tend to be investigated. Furthermore, we demonstrated 4) how and to what extent caregivers are mentioned and involved in the studies. These four points have been driven by our theoretical lens, namely Social Justice-Oriented Design – with a specific focus on the concepts called "designing for recognition" and "designing for accountability" – and Feminist HRI – focusing on power and subject-positioning relations. This framework allowed us to demonstrate the nuances and complexity within this growing field and which stakeholders have been looked at and, ultimately, overlooked.

From those four dimensions, our results indicate that, first and foremost, informal caregivers are not considered key stakeholders within health care in HRI – they were only investigated eight times out of the 52 studies. Yet, this group is still acknowledged as a stakeholder that could ensure the use of robots for the care receivers (they were acknowledged in 24 studies and involved at a low level in five studies). Although, we the authors, refer to this group as informal caregivers, studies do not tend to explicitly identify them as such – including the ones we consider to have investigated informal caregivers. This shows a difficulty in articulating who informal caregivers are in HRI, but the importance of challenging this notion in order to account for them. Furthermore, unlike with formal caregivers or care receivers, informal caregivers were never investigated as a sole group of participants in any study, reinforcing the notion of informal caregivers being ("only") secondary, or even tertiary robot users. This was also viewed within level of involvements of caregivers: even if formal caregivers were not participants, their knowledge was relied on to create relevant robots for care receivers, but this was not the case for informal caregivers. Finally, regarding methods, the least used methods when investigating informal caregivers were those conducted in the wild, interviews and focus groups – which was practically the opposite for studies investigating formal caregivers. In line with this, here are our summarised high-level recommendations:

- 1. Involve informal caregivers more in studies, and this does not necessitate active recruitment: The HRI community should involve informal caregivers at a higher level. This could either be by directly investigating them, or asking informal caregivers for expertise to use in studies, or acknowledging, within the research team, own experiences as informal caregivers to steer studies in health care.
- 2. Recognise the vital role of informal caregivers in HRI, and its possible negative impact: Informal caregivers play a vital role in society, and will also do so in robotic health care applications. Whilst formal caregivers also play such a vital role, they are compensated for it, and we should find ways to ensure that informal caregivers also benefit.
- 3. Policies and guidelines need to recognise, conceptualise and evaluate informal caregivers in HRI: Policies and guidelines regarding HRI in health care need to acknowledge the possible technological reliance on friends and families, and recognise it as a form of informal caregiving that extends across all health care from rehabilitation, to companionship, to monitoring medication.

- 4. Pro-actively leverage on our power within the HRI committee to respond to bottom-up challenges from informal caregivers: The HRI community needs to pro-actively engage in ways to include informal caregivers in studies. As it is a difficult stakeholder to involve, we suggest in the wild studies are probably best suited to get their feedback.
- 5. Create HRI applications specific to informal caregivers: HRI applications should directly look at understanding how to help informal caregivers' well-being, which in turn can improve quality of care for care receivers.

9 Limitations and Further Work

This work was limited to the ACM/IEEE International Conference on Human-Robot Interaction, which only accepts a number of papers only. Ideally this review should include journals to see if there are similar trends. A further limitation is the use of "informal caregiver", these individuals are not solely carers, and who they are caregiving for will also care for them – as the article asking care receivers to throw a birthday party for their caregiver reminds us. Further work could be using this qualitative study and turn it into more quantitative work, to scale up this study to understand how informal caregivers are understood HRI health care studies across more HRI venues and beyond.

Supplementary information. If your article has accompanying supplementary file/s please state so here.

Authors reporting data from electrophoretic gels and blots should supply the full unprocessed scans for key as part of their Supplementary information. This may be requested by the editorial team/s if it is missing.

Please refer to Journal-level guidance for any specific requirements.

Acknowledgments. Acknowledgments are not compulsory. Where included they should be brief. Grant or contribution numbers may be acknowledged.

Please refer to Journal-level guidance for any specific requirements.

Declarations

Some journals require declarations to be submitted in a standardised format. Please check the Instructions for Authors of the journal to which you are submitting to see if you need to complete this section. If yes, your manuscript must contain the following sections under the heading 'Declarations':

- Funding
- Conflict of interest/Competing interests (check journal-specific guidelines for which heading to use)
- Ethics approval
- Consent to participate
- Consent for publication
- Availability of data and materials
- Code availability

• Authors' contributions

If any of the sections are not relevant to your manuscript, please include the heading and write 'Not applicable' for that section.

Editorial Policies for:

Springer journals and proceedings: https://www.springer.com/gp/editorial-policies

Nature Portfolio journals: https://www.nature.com/nature-research/editorial-policies

Scientific Reports: https://www.nature.com/srep/journal-policies/editorial-policies

BMC journals: https://www.biomedcentral.com/getpublished/editorial-policies

Appendix A Section title of first appendix

An appendix contains supplementary information that is not an essential part of the text itself but which may be helpful in providing a more comprehensive understanding of the research problem or it is information that is too cumbersome to be included in the body of the paper.

References

- UNECE: The challenging roles of informal carers. Unece policy brief on ageing no. 20, UNECE (2019)
- [2] Hoffmann, F., Rodrigues, R.: Informal Carers: Who Takes Care of Them. European Centre for Social Welfare Policy and Research Vienna, ??? (2010)
- [3] Grycuk, E., Chen, Y., Almirall-Sanchez, A., Higgins, D., Galvin, M., Kane, J., Kinchin, I., Lawlor, B., Rogan, C., Russell, G., et al.: Care burden, loneliness, and social isolation in caregivers of people with physical and brain health conditions in english-speaking regions: Before and during the covid-19 pandemic. International Journal of Geriatric Psychiatry 37(6) (2022)
- [4] Giesbrecht, M., Crooks, V.A., Williams, A., Hankivsky, O.: Critically examining diversity in end-of-life family caregiving: implications for equitable caregiver support and canada's compassionate care benefit. International Journal for Equity in Health 11(1), 1–13 (2012)
- [5] Caregiving (NAC) & AARP, T.N.A.: Caregiving in the u.s. 2020 report, The National Alliance for Caregiving (NAC) & AARP (2020)
- [6] Zigante, V.: Informal care in europe: Exploring formalisation, availability and quality. Technical report, European Commission & London School of Economics and Political Science (2018)

- [7] Carer Organizations, T.I.A.: Global Carer Facts (2021). https: //internationalcarers.org/carer-facts/global-carer-stats/
- [8] Zhang, Y., Bennett, M., Yeandle, S.: Will i care? the likelihood of being a carer in adult life (2019)
- [9] Royal College of General Practioners: Supporting carers: An action guide for general practitioners and their teams. RCGP, Princess Royal Trust for Carers (2011)
- [10] Deeken, J.F., Taylor, K.L., Mangan, P., Yabroff, K.R., Ingham, J.M.: Care for the caregivers: a review of self-report instruments developed to measure the burden, needs, and quality of life of informal caregivers. Journal of pain and symptom management 26(4), 922–953 (2003)
- [11] Bom, J., Bakx, P., Schut, F., Van Doorslaer, E.: The impact of informal caregiving for older adults on the health of various types of caregivers: a systematic review. The Gerontologist 59(5), 629–642 (2019)
- [12] Borek, A.J., McDonald, B., Fredlund, M., Bjornstad, G., Logan, S., Morris, C.: Healthy parent carers programme: development and feasibility of a novel group-based health-promotion intervention. BMC Public Health 18, 1–16 (2018)
- [13] Baxter, P., Kennedy, J., Senft, E., Lemaignan, S., Belpaeme, T.: From characterising three years of hri to methodology and reporting recommendations. In: 2016 11th Acm/ieee International Conference on Human-robot Interaction (hri), pp. 391–398 (2016). IEEE
- [14] Schrum, M.L., Johnson, M., Ghuy, M., Gombolay, M.C.: Four years in review: Statistical practices of likert scales in human-robot interaction studies. In: Companion of the 2020 ACM/IEEE International Conference on Human-Robot Interaction, pp. 43–52 (2020)
- [15] Zimmerman, M., Bagchi, S., Marvel, J., Nguyen, V.: An analysis of metrics and methods in research from human-robot interaction conferences, 2015–2021. In: 2022 17th ACM/IEEE International Conference on Human-Robot Interaction (HRI), pp. 644–648 (2022). IEEE
- [16] Winkle, K., Lagerstedt, E., Torre, I., Offenwanger, A.: 15 years of (who) man robot interaction: Reviewing the h in human-robot interaction. ACM Transactions on Human-Robot Interaction 12(3), 1–28 (2023)
- [17] Dombrowski, L., Harmon, E., Fox, S.: Social justice-oriented interaction design: Outlining key design strategies and commitments. In: Proceedings of the 2016 ACM Conference on Designing Interactive Systems, pp. 656–671 (2016)
- [18] Winkle, K., McMillan, D., Arnelid, M., Harrison, K., Balaam, M., Johnson, E.,

Leite, I.: Feminist human-robot interaction: Disentangling power, principles and practice for better, more ethical hri. In: Proceedings of the 2023 ACM/IEEE International Conference on Human-Robot Interaction, pp. 72–82 (2023)

- [19] Carduff, E., Finucane, A., Kendall, M., Jarvis, A., Harrison, N., Greenacre, J., Murray, S.A.: Understanding the barriers to identifying carers of people with advanced illness in primary care: triangulating three data sources. BMC family practice 15(1), 1–10 (2014)
- [20] Hengelaar, A.H., Wittenberg, Y., Kwekkeboom, R., Van Hartingsveldt, M., Verdonk, P.: Intersectionality in informal care research: a scoping review. Scandinavian Journal of Public Health 51(1), 106–124 (2023)
- [21] Papadopoulos, C., Lodder, A., Constantinou, G., Randhawa, G.: Systematic review of the relationship between autism stigma and informal caregiver mental health. Journal of autism and developmental disorders 49, 1665–1685 (2019)
- [22] Smyth, C., Blaxland, M., Cass, B.: 'so that's how i found out i was a young carer and that i actually had been a carer most of my life'. identifying and supporting hidden young carers. Journal of Youth Studies 14(2), 145–160 (2011)
- [23] Coe, N.B., Werner, R.M.: Informal caregivers provide considerable front-line support in residential care facilities and nursing homes: Study examines the informal caregivers' considerable frontline support in residential care facilities and nursing homes. Health Affairs 41(1), 105–111 (2022)
- [24] Wormald, A., McGlinchey, E., D'Eath, M., Leroi, I., Lawlor, B., McCallion, P., McCarron, M., O'Sullivan, R., Chen, Y.: Impact of covid-19 pandemic on caregivers of people with an intellectual disability, in comparison to carers of those with other disabilities and with mental health issues: a multicountry study. International Journal of Environmental Research and Public Health 20(4), 3256 (2023)
- [25] Crotty, B.H., Asan, O., Holt, J., Tyszka, J., Erickson, J., Stolley, M., Pezzin, L.E., Nattinger, A.B.: Qualitative assessment of unmet information management needs of informal cancer caregivers: Four themes to inform oncology practice. JCO clinical cancer informatics 4, 521–528 (2020)
- [26] LaValley, S.A., Brady, L.A., Nguyen, M., Jacobs, D.: Family caregivers of older adults: Strategies, tools, and technologies for medication management. Journal of Gerontological Social Work 66(2), 291–302 (2023)
- [27] Shaffer, K.M., Chow, P.I., Cohn, W.F., Ingersoll, K.S., Ritterband, L.M.: Informal caregivers' use of internet-based health resources: an analysis of the health information national trends survey. JMIR aging 1(2), 11051 (2018)
- [28] Messina, A., Amati, R., Albanese, E., Fiordelli, M.: Help-seeking in informal

family caregivers of people with dementia: A qualitative study with isupport as a case in point. International journal of environmental research and public health 19(12), 7504 (2022)

- [29] Schinkinger, S., Tellioğlu, H.: Design implications to systems supporting informal caregivers' daily life. In: Human-Computer Interaction. Applications and Services: 16th International Conference, HCI International 2014, Heraklion, Crete, Greece, June 22-27, 2014, Proceedings, Part III 16, pp. 341–350 (2014). Springer
- [30] Smriti, D.: Bringing emotions into practice: The role of ai in supporting emotional work in informal caregiving. In: Proceedings of the 2014 ACM/IEEE International Conference on Human-robot Interaction. DIS '23 Companion, pp. 37–40, New York, NY, USA (2023). https://doi.org/10.1145/3563703.3593066 . https://doi.org/10.1145/3563703.3593066
- [31] Premanandan, S., Ahmad, A., Cajander, Å., Ågerfalk, P., Gemert-Pijnen, L., et al.: Designing and evaluating it applications for informal caregivers: Protocol for a scoping review. JMIR Research Protocols 12(1), 47650 (2023)
- [32] Chen, S.-C., Jones, C., Moyle, W.: Health professional and workers attitudes towards the use of social robots for older adults in long-term care. International Journal of Social Robotics 12(5), 1135–1147 (2020)
- [33] Louie, W.-Y.G., Li, J., Vaquero, T., Nejat, G.: Socially assistive robots for seniors living in residential care homes: User requirements and impressions. Human-Robot Interactions: Principles, Technologies and Challenges, 75–108 (2015)
- [34] Jeong, S., Breazeal, C., Logan, D., Weinstock, P.: Huggable: impact of embodiment on promoting verbal and physical engagement for young pediatric inpatients. In: 2017 26th IEEE International Symposium on Robot and Human Interactive Communication (RO-MAN), pp. 121–126 (2017). IEEE
- [35] Kim, J.-H., Choi, S.-H., Kim, D., Kim, J., Cho, M.: Animal-robot interaction for pet caring. In: 2009 IEEE International Symposium on Computational Intelligence in Robotics and Automation-(CIRA), pp. 159–164 (2009). IEEE
- [36] Winkle, K., Caleb-Solly, P., Turton, A., Bremner, P.: Social Robots for Engagement in Rehabilitative Therapies: Design Implications from a Study with Therapists (2018). https://doi.org/10.1145/3171221.3171273 . https://doi.org/10. 1145/3171221.3171273https://dl.acm.org/doi/pdf/10.1145/3171221.3171273
- [37] Santos, N.B., Bavaresco, R.S., Tavares, J.E., Ramos, G.d.O., Barbosa, J.L.: A systematic mapping study of robotics in human care. Robotics and Autonomous Systems 144, 103833 (2021)
- [38] Abdi, J., Al-Hindawi, A., Ng, T., Vizcaychipi, M.P.: Scoping review on the use of socially assistive robot technology in elderly care. BMJ open 8(2) (2018)

- [39] Kabacińska, K., Prescott, T.J., Robillard, J.M.: Socially assistive robots as mental health interventions for children: a scoping review. International Journal of Social Robotics 13(5), 919–935 (2021)
- [40] Maibaum, A., Bischof, A., Hergesell, J., Lipp, B.: A critique of robotics in health care. AI & SOCIETY 37(2), 467–477 (2022)
- [41] Pedersen, I., Reid, S., Aspevig, K.: Developing social robots for aging populations: A literature review of recent academic sources. Sociology Compass, 12, e12585 (2018)
- [42] Wright, J.: Robots won't save japan. In: Robots Won't Save Japan. Cornell University Press, ??? (2023)
- [43] Amabili, G., Cucchieri, G., Margaritini, A., Benadduci, M., Barbarossa, F., Luzi, R., Riccardi, G.R., Pelliccioni, G., Maranesi, E., Bevilacqua, R.: Social robotics and dementia: Results from the eware project in supporting older people and their informal caregivers. International Journal of Environmental Research and Public Health 19(20), 13334 (2022)
- [44] Papadopoulos, C., Hill, T., Battistuzzi, L., Castro, N., Nigath, A., Randhawa, G., Merton, L., Kanoria, S., Kamide, H., Chong, N.-Y., et al.: The caresses study protocol: testing and evaluating culturally competent socially assistive robots among older adults residing in long term care homes through a controlled experimental trial. Archives of Public Health 78, 1–10 (2020)
- [45] Laban, G., Morrison, V., Kappas, A., Cross, E.S.: Informal caregivers disclose increasingly more to a social robot over time. In: Chi Conference on Human Factors in Computing Systems Extended Abstracts, pp. 1–7 (2022)
- [46] Burema, D.: A critical analysis of the representations of older adults in the field of human-robot interaction. AI & SOCIETY 37(2), 455–465 (2022)
- [47] Nations, U.: World population prospects 2019. Department of Economic and Social Affairs 141 (2019)
- [48] Buchan, J., Campbell, J., Dhillon, I., Charlesworth, A.: Labour market change and the international mobility of health workers. Health Foundation, 1–37 (2019)
- [49] Dowling, E.: The Care Crisis: What Caused It and How Can We End It? Verso Books, ??? (2022)
- [50] Pu, L., Moyle, W., Jones, C., Todorovic, M.: The effectiveness of social robots for older adults: a systematic review and meta-analysis of randomized controlled studies. The Gerontologist 59(1), 37–51 (2019)

- [51] Winkle, K., Melsión, G.I., McMillan, D., Leite, I.: Boosting robot credibility and challenging gender norms in responding to abusive behaviour: A case for feminist robots. In: Companion of the 2021 ACM/IEEE International Conference on Human-robot Interaction, pp. 29–37 (2021)
- [52] Perugia, G., Guidi, S., Bicchi, M., Parlangeli, O.: The shape of our bias: Perceived age and gender in the humanoid robots of the abot database. In: Proceedings of the 2022 ACM/IEEE International Conference on Human-Robot Interaction, pp. 110–119 (2022)
- [53] Moradbakhti, L., Mara, M., Castellano, G., Winkle, K.: (counter-) stereotypical gendering of robots in care: Impact on needs satisfaction and gender role concepts in men and women users. International Journal of Social Robotics 15(11), 1769–1790 (2023)
- [54] Seaborn, K., Barbareschi, G., Chandra, S.: Not only weird but "uncanny"? a systematic review of diversity in human–robot interaction research. International Journal of Social Robotics, 1–30 (2023)
- [55] Papadopoulos, C., Castro, N., Nigath, A., Davidson, R., Faulkes, N., Menicatti, R., Khaliq, A.A., Recchiuto, C., Battistuzzi, L., Randhawa, G., et al.: The caresses randomised controlled trial: exploring the health-related impact of culturally competent artificial intelligence embedded into socially assistive robots and tested in older adult care homes. International Journal of Social Robotics 14(1), 245–256 (2022)
- [56] Page, M.J., McKenzie, J.E., Bossuyt, P.M., Boutron, I., Hoffmann, T.C., Mulrow, C.D., Shamseer, L., Tetzlaff, J.M., Akl, E.A., Brennan, S.E., *et al.*: The prisma 2020 statement: an updated guideline for reporting systematic reviews. Systematic reviews **10**(1), 1–11 (2021)
- [57] Braun, V., Clarke, V.: Using thematic analysis in psychology. Qualitative research in psychology 3(2), 77–101 (2006)
- [58] Kraft, K., Smart, W.D.: Seeing is Comforting: Effects of Teleoperator Visibility in Robot-Mediated Health Care. IEEE Press (2016)
- [59] O'Brien, C., O'Mara, M., Issartel, J., McGinn, C.: Exploring the Design Space of Therapeutic Robot Companions for Children (2021). https://doi. org/10.1145/3434073.3444669 . https://doi.org/10.1145/3434073.3444669https: //dl.acm.org/doi/pdf/10.1145/3434073.3444669
- [60] Nanavati, A., Alves-Oliveira, P., Schrenk, T., Gordon, E.K., Cakmak, M., Srinivasa, S.S.: Design Principles for Robot-Assisted Feeding in Social Contexts (2023). https://doi.org/10.1145/3568162.3576988 . https://doi.org/10. 1145/3568162.3576988https://dl.acm.org/doi/pdf/10.1145/3568162.3576988

- [61] Moorman, N., Hedlund-Botti, E., Schrum, M., Natarajan, M., Gombolay, M.C.: Impacts of Robot Learning on User Attitude and Behavior (2023). https://doi. org/10.1145/3568162.3576996 . https://doi.org/10.1145/3568162.3576996https: //dl.acm.org/doi/pdf/10.1145/3568162.3576996
- [62] Zhang, F., Broz, F., Dertien, E., Kousi, N., Gurp, J.A.M.v., Ferrari, O.I., Malagon, I., Barakova, E.I.: Understanding Design Preferences for Robots for Pain Management: A Co-Design Study. IEEE Press (2022)
- [63] Gallenberger, D., Bhattacharjee, T., Kim, Y., Srinivasa, S.S.: Transfer depends on acquisition: analyzing manipulation strategies for robotic feeding. IEEE Press (2020)
- [64] Lee, H.R., Sun, F., Iqbal, T., Roberts, B.: Reimagining Robots for Dementia: From Robots for Care-receivers/giver to Robots for Carepartners (2023). https://doi.org/10.1145/3568162.3578624 . https://doi.org/10.1145/3568162. 3578624https://dl.acm.org/doi/pdf/10.1145/3568162.3578624
- [65] Birmingham, C., Perez, A., Matarić, M.: Perceptions of Cognitive and Affective Empathetic Statements by Socially Assistive Robots. IEEE Press (2022)
- [66] Drift, E.J.G., Beun, R.-J., Looije, R., Henkemans, O.A.B., Neerincx, M.A.: A remote social robot to motivate and support diabetic children in keeping a diary (2014). https://doi.org/10.1145/2559636.2559664 . https://doi.org/10. 1145/2559636.2559664https://dl.acm.org/doi/pdf/10.1145/2559636.2559664
- [67] Chevalier, P., Raiola, G., Martin, J.-C., Isableu, B., Bazile, C., Tapus, A.: Do Sensory Preferences of Children with Autism Impact an Imitation Task with a Robot? (2017). https://doi.org/10.1145/2909824.3020234 . https://doi.org/10. 1145/2909824.3020234https://dl.acm.org/doi/pdf/10.1145/2909824.3020234
- [68] Caine, K., Šabanovic, S., Carter, M.: The effect of monitoring by cameras and robots on the privacy enhancing behaviors of older adults (2012). https://doi. org/10.1145/2157689.2157807 . https://doi.org/10.1145/2157689.2157807https: //dl.acm.org/doi/pdf/10.1145/2157689.2157807
- [69] Winkle, K., Lemaignan, S., Caleb-Solly, P., Leonards, U., Turton, A., Bremner, P.: Effective persuasion strategies for socially assistive robots. IEEE Press (2020)
- [70] Tsui, K., Yanco, H., Kontak, D., Beliveau, L.: Development and evaluation of a flexible interface for a wheelchair mounted robotic arm (2008). https://doi. org/10.1145/1349822.1349837 . https://doi.org/10.1145/1349822.1349837https: //dl.acm.org/doi/pdf/10.1145/1349822.1349837
- [71] Beer, J.M., Smarr, C.-A., Chen, T.L., Prakash, A., Mitzner, T.L., Kemp, C.C., Rogers, W.A.: The domesticated robot: design guidelines for assisting older adults to age in place (2012). https://doi.org/10.

 $\frac{1145}{2157689.2157806} \quad https://doi.org/10.1145/2157689.2157806 https://dl.acm.org/doi/pdf/10.1145/2157689.2157806$

- [72] Beer, J.M., Takayama, L.: Mobile remote presence systems for older adults: acceptance, benefits, and concerns (2011). https://doi.org/10. 1145/1957656.1957665 . https://doi.org/10.1145/1957656.1957665https://dl. acm.org/doi/pdf/10.1145/1957656.1957665
- [73] Rea, D.J., Schneider, S., Kanda, T.: "Is this all you can do? Harder!": The Effects of (Im)Polite Robot Encouragement on Exercise Effort (2021). https://doi.org/ 10.1145/3434073.3444660 . https://doi.org/10.1145/3434073.3444660https://dl. acm.org/doi/pdf/10.1145/3434073.3444660
- [74] Ranganeni, V., Sinclair, M., Ofek, E., Miller, A., Campbell, J., Kolobov, A., Cutrell, E.: Exploring Levels of Control for a Navigation Assistant for Blind Travelers (2023). https://doi.org/10.1145/3568162.3578630 . https://doi.org/10. 1145/3568162.3578630https://dl.acm.org/doi/pdf/10.1145/3568162.3578630
- [75] Bhattacharjee, T., Gordon, E.K., Scalise, R., Cabrera, M.E., Caspi, A., Cakmak, M., Srinivasa, S.S.: Is More Autonomy Always Better? Exploring Preferences of Users with Mobility Impairments in Robot-assisted Feeding (2020). https://doi. org/10.1145/3319502.3374818 . https://doi.org/10.1145/3319502.3374818https: //dl.acm.org/doi/pdf/10.1145/3319502.3374818
- [76] Wilson, R., Keane, I., Jones, R.: Affective Responses of Older Adults to the Anthropomorphic GenieConnect Companion Robot During Lockdown of the COVID19 Pandemic. IEEE Press (2022)
- [77] Briggs, P., Scheutz, M., Tickle-Degnen, L.: Are Robots Ready for Administering Health Status Surveys': First Results from an HRI Study with Subjects with Parkinson's Disease (2015). https://doi.org/10.1145/2696454.2696476 . https://doi.org/10.1145/2696454.2696476https://dl.acm.org/doi/pdf/10.1145/ 2696454.2696476
- [78] Heerink, M., Kröse, B., Wielinga, B., Evers, V.: Enjoyment intention to use and actual use of a conversational robot by elderly people (2008). https://doi. org/10.1145/1349822.1349838 . https://doi.org/10.1145/1349822.1349838https: //dl.acm.org/doi/pdf/10.1145/1349822.1349838
- [79] Prakash, A., Beer, J.M., Deyle, T., Smarr, C.-A., Chen, T.L., Mitzner, T.L., Kemp, C.C., Rogers, W.A.: Older adults' medication management in the home: how can robots help? IEEE Press (2013)
- [80] Chen, T.L., Kemp, C.C.: Lead me by the hand: evaluation of a direct physical interface for nursing assistant robots. IEEE Press (2010)
- [81] Matsumoto, S., Ghosh, P., Jamshad, R., Riek, L.D.: Robot, Uninterrupted:

 $\label{eq:care_constraint} \begin{array}{l} \mbox{Telemedical Robots to Mitigate Care Disruption (2023). https://doi.org/10.1145/3568162.3576994.https://doi.org/10.1145/3568162.3576994https://dl. acm.org/doi/pdf/10.1145/3568162.3576994 \end{array}$

- [82] Camilleri, A., Dogramadzi, S., Caleb-Solly, P.: Learning from Carers to inform the Design of Safe Physically Assistive Robots - Insights from a Focus Group Study. IEEE Press (2022)
- [84] Lippi, V., Filippeschi, A., Camardella, C., Porcini, F., Maurer, C., Lencioni, L.: EXOSMOOTH: Test of Innovative EXOskeleton Control for SMOOTH Assistance, With and Without Ankle Actuation. IEEE Press (2022)
- [85] Novak, D., Nagle, A., Riener, R.: Can two-player games increase motivation in rehabilitation robotics? (2014)
- [86] Hoffman, G., Birnbaum, G.E., Vanunu, K., Sass, O., Reis, H.T.: Robot responsiveness to human disclosure affects social impression and appeal (2014). https://doi.org/10.1145/2559636.2559660 . https://doi.org/10.1145/ 2559636.2559660https://dl.acm.org/doi/pdf/10.1145/2559636.2559660
- [87] Chen, T.L., King, C.-H., Thomaz, A.L., Kemp, C.C.: Touched by a robot: an investigation of subjective responses to robot-initiated touch (2011).https://doi.org/10.1145/1957656.1957818 .https://doi.org/10.1145/1957656.1957818https://dl.acm.org/doi/pdf/10.1145/1957656.1957818
- [88] Neerincx, A., Leven, J., Wolfert, P., Graaf, M.M.A.d.: The Effect of Simple Emotional Gesturing in a Socially Assistive Robot on Child's Engagement at a Group Vaccination Day (2023). https://doi.org/10. 1145/3568162.3576960 . https://doi.org/10.1145/3568162.3576960https://dl. acm.org/doi/pdf/10.1145/3568162.3576960
- [89] Odabasi, C., Graf, F., Lindermayr, J., Patel, M., Baumgarten, S.D., Graf, B.: Refilling Water Bottles in Elderly Care Homes With the Help of a Safe Service Robot (2022)
- [90] Tanqueray, L., Paulsson, T., Zhong, M., Larsson, S., Castellano, G.: Gender Fairness in Social Robotics: Exploring a Future Care of Peripartum Depression. IEEE Press (2022)
- [91] Kubota, A., Pei, R., Sun, E., Cruz-Sandoval, D., Kim, S., Riek, L.D.: Get SMART: Collaborative Goal Setting with Cognitively Assistive Robots (2023). https://doi.org/10.1145/3568162.3576993 . https://doi.org/10.1145/

3568162.3576993https://dl.acm.org/doi/pdf/10.1145/3568162.3576993

- [92] Spitale, M., Axelsson, M., Gunes, H.: Robotic Mental Well-being Coaches for the Workplace: An In-the-Wild Study on Form (2023). https://doi.org/ 10.1145/3568162.3577003 . https://doi.org/10.1145/3568162.3577003https://dl. acm.org/doi/pdf/10.1145/3568162.3577003
- [93] Kubota, A., Cruz-Sandoval, D., Kim, S., Twamley, E.W., Riek, L.D.: Cognitively Assistive Robots at Home: HRI Design Patterns for Translational Science. IEEE Press (2022)
- [94] Cruz-Sandoval, D., Morales-Tellez, A., Sandoval, E.B., Favela, J.: A Social Robot as Therapy Facilitator in Interventions to Deal with Dementia-related Behavioral Symptoms (2020). https://doi.org/10.1145/3319502.3374840 . https://doi.org/10.1145/3319502.3374840https://dl.acm.org/doi/pdf/10.1145/ 3319502.3374840
- [95] Feingold-Polak, R., Levy-Tzedek, S.: Social Robot for Rehabilitation: Expert Clinicians and Post-Stroke Patients' Evaluation Following a Long-Term Intervention (2020). https://doi.org/10.1145/3319502.3374797 . https://doi.org/10. 1145/3319502.3374797https://dl.acm.org/doi/pdf/10.1145/3319502.3374797
- [96] Ozgur, A.G., Wessel, M.J., Johal, W., Sharma, K., Özgür, A., Vuadens, P., Mondada, F., Hummel, F.C., Dillenbourg, P.: Iterative Design of an Upper Limb Rehabilitation Game with Tangible Robots (2018). https://doi.org/ 10.1145/3171221.3171262.https://doi.org/10.1145/3171221.3171262https://dl. acm.org/doi/pdf/10.1145/3171221.3171262
- [97] Lee, H.R., Šabanović, S., Chang, W.-L., Nagata, S., Piatt, J., Bennett, C., Hakken, D.: Steps Toward Participatory Design of Social Robots: Mutual Learning with Older Adults with Depression (2017). https://doi.org/ 10.1145/2909824.3020237 . https://doi.org/10.1145/2909824.3020237 https://dl. acm.org/doi/pdf/10.1145/2909824.3020237
- [98] Hebesberger, D., Dondrup, C., Koertner, T., Gisinger, C., Pripfl, J.: Lessons Learned from the Deployment of a Long-term Autonomous Robot as Companion in Physical Therapy for Older Adults with Dementia: A Mixed Methods Study. IEEE Press (2016)
- [99] Moosaei, M., Das, S.K., Popa, D.O., Riek, L.D.: Using Facially Expressive Robots to Calibrate Clinical Pain Perception (2017). https://doi.org/ 10.1145/2909824.3020216 . https://doi.org/10.1145/2909824.3020216https://dl. acm.org/doi/pdf/10.1145/2909824.3020216
- [100] Park, C.H., Howard, A.M.: Real world haptic exploration for telepresence of the visually impaired (2012). https://doi.org/10.1145/2157689.2157701 . https://doi.org/10.1145/2157689.2157701https://dl.acm.org/doi/pdf/10.1145/
 - 39

2157689.2157701

- [101] Tsoi, N., Connolly, J., Adéníran, E., Hansen, A., Pineda, K.T., Adamson, T., Thompson, S., Ramnauth, R., Vázquez, M., Scassellati, B.: Challenges Deploying Robots During a Pandemic: An Effort to Fight Social Isolation Among Children (2021). https://doi.org/10.1145/3434073.3444665 . https://doi.org/10. 1145/3434073.3444665https://dl.acm.org/doi/pdf/10.1145/3434073.3444665
- [102] Shi, Z., Chen, H., Velentza, A.-M., Liu, S., Dennler, N., O'Connell, A., Mataric, M.: Evaluating and Personalizing User-Perceived Quality of Text-to-Speech Voices for Delivering Mindfulness Meditation with Different Physical Embodiments (2023). https://doi.org/10.1145/3568162.3576987 . https://doi.org/10. 1145/3568162.3576987
- [103] Fang, N., Zhang, C., Sankaran, S., Ren, S.: Role of Socially Assistive Robots in Reducing Anxiety and Preserving Autonomy in Children. IEEE Press (2022)
- [104] Moharana, S., Panduro, A.E., Lee, H.R., Riek, L.D.: Robots for joy, robots for sorrow: community based robot design for dementia caregivers. IEEE Press (2020)
- [105] Herlant, L.V., Holladay, R.M., Srinivasa, S.S.: Assistive Teleoperation of Robot Arms via Automatic Time-Optimal Mode Switching. IEEE Press (2016)
- [106] Antony, V.N., Cho, S.M., Huang, C.-M.: Co-Designing with Older Adults, for Older Adults: Robots to Promote Physical Activity (2023). https://doi.org/ 10.1145/3568162.3576995 . https://doi.org/10.1145/3568162.3576995https://dl. acm.org/doi/pdf/10.1145/3568162.3576995
- [107] Caleb-Solly, P., Dogramadzi, S., Ellender, D., Fear, T., Heuvel, H.v.d.: A mixed-method approach to evoke creative and holistic thinking about robots in a home environment (2014). https://doi.org/10.1145/2559636.2559681 . https://doi.org/10.1145/2559636.2559681https://dl.acm.org/doi/pdf/10.1145/ 2559636.2559681
- [108] Valencia, S., Luria, M., Pavel, A., Bigham, J.P., Admoni, H.: Co-designing Socially Assistive Sidekicks for Motion-based AAC (2021). https://doi.org/ 10.1145/3434073.3444646 . https://doi.org/10.1145/3434073.3444646https://dl. acm.org/doi/pdf/10.1145/3434073.3444646
- [109] Tanqueray, L., Paulsson, T., Zhong, M., Larsson, S., Castellano, G.: Gender fairness in social robotics: Exploring a future care of peripartum depression. In: 2022 ACM/IEEE International Conference on Human-Robot Interaction. IEEE, ??? (2022)
- [110] Harding, S.G.: The Feminist Standpoint Theory Reader: Intellectual and Political Controversies. Psychology Press, ??? (2004)

- [111] Harrison, K., Johnson, E.: Affective corners as a problematic for design interactions. ACM Transactions on Human-Robot Interaction (2023)
- [112] Thunberg, S., Johnson, E., Ziemke, T.: Investigating healthcare workers' technostress when welfare technology is introduced in long-term care facilities. Behaviour & Information Technology, 1–13 (2023)